\input{ThisWeek.tex}
\newcommand{\classday}{\Thursday}
\input{../TemplateWeek/InClassWork.tex}
\input{Exercises.tex}

%\booltrue{teacher}

%Note that I added some stuff to deal with overflow from previous days.  We were supposed to start with the 3.1 exercises.

%Ultimately everyone got most of the way though 4, though they needed increasing amounts of help to get through the algebra.  It might save some time to offer some hints there.


% \section*{Warm Up}

% Suppose you want to find $\lim_{x\rightarrow\infty}\frac{f(x)}{g(x)}$.  What conditions do you need in order to apply the limit law for division?


\section*{Exercise 1 (from Section 2.4)}
\textit{If your group finished this problem on Tuesday, skip to the next one.}

\Exercise{E2-4_Main7}

\section*{Exercise 2 (from Section 2.4)}
\Exercise{E2-4_Main8}


\section*{Exercise 3}
\Exercise{E3-1_Main1}

\section*{Exercise 4}
\Exercise{E3-1_Main2}

\section*{Exercise 5}
\Exercise{E3-1_Main3}

\section*{Exercise 6}
\Exercise{E3-1_Main4}

%\section*{Exercise 7}
%\Exercise{E3-1_Main5}

\section*{Exercise 8}
\Exercise{E3-1_Main6}

\end{document}

